

## **Geophysical Research Letters**<sup>•</sup>

### **RESEARCH LETTER**

10.1029/2021GL094566

#### **Key Points:**

- An ensemble of ice-ocean simulations is used to study conditions in the Amundsen Sea over the twentieth century
- The simulations show a longterm ocean warming trend on the continental shelf, which is not yet clear in the shorter observational record
- The trends are partly driven by anthropogenic forcing, suggesting the ocean and ice shelves may also be sensitive to future climate policy

#### Supporting Information:

Supporting Information may be found in the online version of this article.

#### Correspondence to:

K. A. Naughten, kaight@bas.ac.uk

#### Citation:

Naughten, K. A., Holland, P. R., Dutrieux, P., Kimura, S., Bett, D. T., & Jenkins, A. (2022). Simulated twentiethcentury ocean warming in the Amundsen Sea, West Antarctica. *Geophysical Research Letters*, 49, e2021GL094566. https://doi.org/10.1029/2021GL094566

Received 2 JUN 2021 Accepted 14 FEB 2022

#### **Author Contributions:**

Conceptualization: Kaitlin A. Naughten, Paul R. Holland, Adrian Jenkins Formal analysis: Kaitlin A. Naughten Funding acquisition: Paul R. Holland, Satoshi Kimura, Adrian Jenkins Investigation: Kaitlin A. Naughten Methodology: Kaitlin A. Naughten, Paul R. Holland

Project Administration: Paul R. Holland Software: Kaitlin A. Naughten, Paul R. Holland, Satoshi Kimura, David T. Bett Supervision: Paul R. Holland Validation: Kaitlin A. Naughten, Paul R. Holland, Pierre Dutrieux

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Simulated Twentieth-Century Ocean Warming in the Amundsen Sea, West Antarctica

Kaitlin A. Naughten<sup>1</sup> <sup>(1</sup>), Paul R. Holland<sup>1</sup>, Pierre Dutrieux<sup>1</sup> <sup>(1</sup>), Satoshi Kimura<sup>2</sup> <sup>(1</sup>), David T. Bett<sup>1</sup> <sup>(1</sup>), and Adrian Jenkins<sup>3</sup> <sup>(1</sup>)

<sup>1</sup>British Antarctic Survey, Cambridge, UK, <sup>2</sup>Japan Agency for Marine-Earth Science and Technology, Yokosuka, Japan, <sup>3</sup>Department of Geography and Environmental Sciences, Northumbria University, Newcastle upon Tyne, UK

**Abstract** Rapid ice loss is occurring in the Amundsen Sea sector of the West Antarctic Ice Sheet. This ice loss is assumed to be a long-term response to oceanographic forcing, but ocean conditions in the Amundsen Sea are unknown prior to 1994. Here we present a modeling study of Amundsen Sea conditions from 1920 to 2013, using an ensemble of ice-ocean simulations forced by climate model experiments. We find that during the early twentieth century, the Amundsen Sea likely experienced more sustained cool periods than at present. Warm periods become more dominant over the simulations (mean trend 0.33°C/century) causing an increase in ice shelf melting. The warming is likely driven by an eastward wind trend over the continental shelf break that is partly anthropogenically forced. Our simulations suggest that the Amundsen Sea responded to historical greenhouse gas forcing, and that future changes in emissions are also likely to affect the region.

**Plain Language Summary** Ice loss from the West Antarctic Ice Sheet is one of the fastest growing contributions to sea level rise. The pattern of melting suggests that the ocean has been warming in the Amundsen Sea, but we don't know what the ocean was like in this region before observations started in 1994. In this study, we model the ocean conditions over the last century, using a set of 20 ice-ocean simulations fed with the output of larger climate models. Our simulations suggest that the Amundsen Sea used to be cooler than it is today. Over the last century, ocean temperatures have increased, but this is hard to detect in observations because the trend is obscured by frequent swings between cold and warm conditions. The main cause of the warming in our simulations is changes in the winds near the edge of the Amundsen Sea that pump more warm water from the deep ocean onto the shallower continental shelf. These wind trends are partly caused by greenhouse gas emissions from the burning of fossil fuels. Whether the Amundsen Sea continues to warm, or whether temperatures stabilize, will likely hinge on the success of future climate policies to reduce emissions.

#### 1. Introduction

The West Antarctic Ice Sheet (WAIS) is a region of particular concern for global sea level rise. Ice loss from the WAIS is mainly controlled by the ocean, which melts the floating ice shelves fringing the ice sheet. Thinning of these ice shelves causes a loss of buttressing and acceleration of the upstream glaciers, which then unpin from the bedrock and cause grounding line retreat. The geometry of the WAIS makes it more susceptible to destabilization: It is largely grounded below sea level (Fretwell et al., 2013) with many inland-sloping regions known as retrograde beds. Depending on the level of ice shelf buttressing (Gudmundsson et al., 2012), retrograde beds could lead to an unstable regime, in which grounding line retreat would continue even without further warming of the ocean (Schoof, 2007).

It is clear from observations that ice loss has already begun. Satellite estimates of mass loss from the Antarctic Ice Sheet show that the WAIS is the main contributor,  $losing \approx 160$  Gt of ice per year (Shepherd et al., 2018). This ice loss is focused in the Amundsen and Bellingshausen Seas (Paolo et al., 2015), suggesting ocean-driven melting is the cause. Rapid grounding line retreat has also been observed in the Amundsen Sea, and in many regions the grounding lines have crossed into retrograde beds (Rignot et al., 2014). There is evidence that this period of ice loss dates back to at least the 1970s (Jenkins et al., 2010), and perhaps as early as the 1940s (Smith et al., 2017).

Given the strong evidence of increased ocean-driven melting in the Amundsen Sea, it may seem surprising that there is no clear warming signal in ocean temperatures on the continental shelf. Rather, the water at middepth, which can most easily access the ice shelf cavities, oscillates between cold Winter Water (WW,  $\approx -1^{\circ}$ C) and warm Circumpolar Deep Water (CDW,  $0-1^{\circ}$ C). This oscillation occurs on a roughly decadal cycle (Dutrieux



Naughten

Adrian Jenkins

Visualization: Kaitlin A. Naughten

Writing - original draft: Kaitlin A.

Writing – review & editing: Kaitlin A. Naughten, Paul R. Holland, Pierre

Dutrieux, Satoshi Kimura, David T. Bett,

et al., 2014; Jenkins et al., 2018) as the thermocline, which separates the two water masses, rises and falls. As a result, observations are so dominated by decadal variability that trends based on the short records available cannot be relied upon (Schmidtko et al., 2014). It is possible that ice loss from the WAIS is driven by geometric feedbacks (such as self-sustained grounding line retreat on a retrograde bed), triggered by a decadal anomaly earlier in the twentieth century and largely unaffected by recent temperatures (Jenkins et al., 2016). Alternatively, variability in subsurface temperatures could be obscuring a longer-term signal by which warm periods become more dominant.

A major driver of Amundsen Sea variability is the winds over the nearby continental shelf break (Jenkins et al., 2016; Thoma et al., 2008). Eastward wind anomalies favor the onshore transport of warm CDW, which raises the thermocline and warms the shelf at middepth. Conversely, westward wind anomalies reduce onshore transport, leading to a deeper thermocline with cold WW at middepth. Winds at the shelf break are strongly influenced by Pacific variability, with eastward anomalies associated with positive phases of El Niño Southern Oscillation (ENSO) and the Interdecadal Pacific Oscillation (IPO). However, climate model simulations suggest these cycles are superimposed on a long-term eastward trend in winds over the shelf break (P. R. Holland et al., 2019). This trend results partly from the ongoing strengthening and poleward shift of the Southern Hemisphere westerly winds, a feature of anthropogenic greenhouse gas forcing (Swart et al., 2015). It would be expected to cause long-term warming of the Amundsen Sea, manifested as an increased presence of decadal warm anomalies.

While the link between eastward wind anomalies and increased ice shelf melting offers a credible explanation for WAIS ice loss, it cannot be thoroughly tested based on observations alone. In-situ oceanographic measurements in the Amundsen Sea have only existed since 1994 (S. S. Jacobs & Hellmer, 1996), and their spatial and temporal coverage is still sparse. As such, we have partially observed only two cycles of the continental shelf's response to decadal wind forcing, which is insufficient to resolve centennial trends. It is therefore unclear whether the relationship between winds and melting holds on timescales longer than decadal. Negative feedbacks on the shelf, related to changes in baroclinic ocean circulation or surface fluxes, could plausibly counteract a centennial trend in wind forcing. It is therefore unknown what conditions the Amundsen Sea experienced earlier in the twentieth century, and whether warm periods have indeed become more common as a result of human activity.

In this study we use an ice-ocean model of the Amundsen Sea, forced by climate model reconstructions beginning in 1920, to fill these gaps in our knowledge. By running an ensemble of centennial simulations, we can detect long-term trends in ocean conditions on the continental shelf. In particular, we evaluate whether the ocean responds to changes in atmospheric forcing over the last century. This result will determine, by extension, whether the Amundsen Sea is also likely to respond to future changes in greenhouse gas emissions.

#### 2. Methodology

Our modeling approach is to perform a hypothesis test regarding the role of historical atmospheric forcing on the Amundsen Sea, rather than to create a complete reconstruction or hindcast. We do not consider changes in non-atmospheric boundary conditions, including ice shelf geometry, iceberg calving, and the lateral ocean boundaries. These variables, often poorly constrained by observations at present, are almost completely unconstrained in the past. While the Amundsen Sea sector is known to have experienced grounding line retreat (Jenkins et al., 2010) and changes in calving (MacGregor et al., 2012) over the past decades, it is not clear how these changes could best be prescribed. Furthermore, global climate models are not designed to simulate water masses accurately on the Antarctic continental shelf (Asay-Davis et al., 2017), so their historical ocean simulations cannot be relied upon in this region. We therefore keep all boundary conditions, apart from atmospheric forcing, fixed at the present day. This design means we do not consider the effects of remote ocean processes (Spence et al., 2017), ice shelf geometric feedbacks (Donat-Magnin et al., 2017), or changes in the origin and properties of CDW (Nakayama et al., 2018; Sallée et al., 2013).

Our simulations are performed using the Massachusetts Institute of Technology general circulation model (MITgcm) (Losch, 2008; Marshall et al., 1997) including components for the ocean, sea ice, and ice shelf thermodynamics. Here we build on the Amundsen Sea configuration of Assmann et al. (2013), Kimura et al. (2017), and Bett et al. (2020), which has been updated and re-tuned to provide best agreement with observations when forced with the latest ERA5 atmospheric reanalysis (Copernicus Climate Change Service, 2017). A detailed model description is given in Text S1 in the Supporting Information S1.

When forced with ERA5, the model generally agrees with observations, in terms of mean conditions as well as interannual variability (Text S2 in the Supporting Information S1). Its main bias is a predisposition to occasional convection on the continental shelf, which temporarily cools and freshens the deep waters. Such an event occurs once during the ERA5-forced simulation, although it was not observed during this time. The potential impact of this bias on our results is discussed later.

In this study, we present twentieth-century simulations forced with the atmospheric output of the Pacific Pacemaker Ensemble (PACE) of climate model simulations (Schneider & Deser, 2018). PACE consists of 20 realizations of the global Community Earth System Model (CESM), running from 1920 to 2013 with historical forcing including changes in greenhouse gases. The simulations are also constrained to follow observed sea surface temperature anomalies in the tropical Pacific. This constraint is useful for studying the last century in the Amundsen Sea, as Pacific variability has a strong influence on this region, and has contributed to centennial trends (P. R. Holland et al., 2019). These simulations preserve the observed record of Pacific variability, including ENSO and the IPO, while sampling the range of other modes of variability at all timescales. Since PACE has a cold and dry bias over Antarctica, we apply spatially varying bias corrections to some variables (Text S3 in the Supporting Information S1), but not to winds over the shelf break. Each MITgcm simulation is preceded by a 30-year spin-up, consisting of the forcing period 1920–1949 repeated once. Drift occurs in approximately the first 10 years of spinup and not thereafter.

#### 3. Results

#### 3.1. Simulated Trends Over the Last Century

It has previously been shown (P. R. Holland et al., 2019) that the PACE ensemble exhibits a significant eastward trend in zonal wind over the Amundsen Sea shelf break (Figure 1a). Our ice-ocean simulations show, for the first time, that these wind trends correspond with oceanographic changes on the Amundsen Sea continental shelf. We find a significant warming trend at middepth (Figure 1b) of  $0.33^{\circ}$ C/century in the ensemble mean. We also find a significant increase in ice shelf melting (Figure 1c) of 47%/century, although this trend may be less reliable as we do not consider ice shelf geometric feedbacks. In both cases, these trends represent an increase of 1–2 standard deviations over the simulation. Middepth salinity on the continental shelf also has a significant positive trend of 0.021 psu/century, consistent with greater influx of CDW across the shelf break. (The significance of ensemble mean trends in this study is determined using a 2-sided *t*-test for the 20 individual trends, against the null hypothesis that the mean is zero.)

The ensemble mean trend represents the response of the system to both radiative forcing and centennial-scale Pacific variability (Schneider & Deser, 2018). The detrended ensemble mean variability (variations in the dark blue line relative to the black trend line in Figure 1) represents shorter-term Pacific variability. The timing of decadal variations in the ensemble mean is largely consistent with the glaciological record. We simulate a warm period around 1970, with reduced ensemble spread indicating high confidence. This warm period aligns with the final ungrounding of Pine Island Glacier from a submarine ridge, as estimated from sediment records (Smith et al., 2017) and remote sensing (Jenkins et al., 2010; shaded region 2 in Figure 1c). We also simulate warm periods in the early 1990s and early 2000s, corresponding with the onset of recent thinning of the Pine Island and Thwaites Glaciers respectively (Konrad et al., 2017; shaded regions 3 and 4 in Figure 1c). However, our simulations do not show an obvious warm period in the 1940s, which is estimated as the initial ungrounding of Pine Island ergion 1 in Figure 1c). While the strong El Niño of the early 1940s (Schneider & Steig, 2008; Steig et al., 2013) is apparent in the zonal winds, it is less obvious in the temperature and melting.

The spread of the ensemble indicates internal non-Pacific variability, including Atlantic and Indian Ocean modes of variability, and local feedbacks on the continental shelf. This ensemble spread is substantial, demonstrating that tropical Pacific variability is far from the only determinant of conditions on the continental shelf. The ERA5-forced simulation is shown in Figure 1 for comparison, and is equivalent to an individual ensemble member. It generally falls within the ensemble spread, suggesting our PACE bias corrections are successful. Hovmöller





**Figure 1.** Timeseries of conditions in the Pacific Pacemaker Ensemble (PACE) simulations (blue) and the ERA5 simulation (red). Thinner blue lines show the 20 PACE ensemble members, while the thicker blue line is the ensemble mean. (a) Zonal wind (m/s) averaged over the shelf break. (b) Temperature (°C) averaged over the Amundsen Sea continental shelf between 200 and 700 m. (c) Basal melt flux (Gt/y) for ice shelves between Dotson and Cosgrove inclusive (see Figure 2). The regions used for (a and b) are defined in Figure S1 in the Supporting Information S1. All timeseries are 2-year running means. Also shown are the ensemble mean trends (black lines and printed values), all of which are significant at the 99% level. For (c), the trend is also expressed as a percentage of the ensemble mean over the first 30 years. The second y-axis on the right of each panel scales the results to show the anomaly, in standard deviations, from the first 30 years. Shaded time periods show the estimated dates of major glaciological events, labeled in (c): (1, 2) the initial and final ungrounding of Pine Island Glacier from a submarine ridge (Smith et al., 2017); (3, 4) the onset of recent thinning of Pine Island and Thwaites Glaciers, respectively (Korrad et al., 2017).





## Trends in horizontal heat transport at 400m (kW/m<sup>2</sup>/century)

Figure 2. Ensemble mean trends in horizontal advection of heat, relative to the surface freezing point. The x and y components of the trend are calculated separately using annual means, and their magnitude is shaded in red. Vectors are plotted where the magnitude exceeds 125 kW/m<sup>2</sup>/century. Trends which are not significant at the 95% level are set to zero; this applies to the x and y components individually. The dotted blue line shows the 1000 m depth contour on the shelf break. Blue labels indicate the ice shelves (G = Getz, D = Dotson, Cr = Crosson, T = Thwaites, P = Pine Island, Co = Cosgrove, A = Abbot, V = Venable) and other topographic features (DG = Dotson-Getz Trough, PITW = Pine Island-Thwaites West Trough, PITE = Pine Island-Thwaites East Trough, BR = Bear Ridge, PIB = Pine Island Bay).

temperature plots (Figure S8 in the Supporting Information S1) confirm that the PACE simulations resemble ERA5, although the phasing of the variability is heterogeneous.

#### 3.2. Mechanism of Warming

Coincident trends in winds and middepth temperature are not proof of causation. However, a closer analysis of the warming reveals behavior consistent with the changes in winds. Trends in horizontal heat transport near the base of the thermocline (Figure 2) show a strengthening of existing pathways of onshore CDW flux. Increased eastward heat transport along the shelf break at 400 m depth indicates a strengthening of the undercurrent (Walker et al., 2013), just as would be expected from eastward wind anomalies (Thoma et al., 2008). The undercurrent then turns onshore through three main bathymetric troughs and flows toward the cavities. Note that strengthened heat transport could also be influenced by the ice shelf meltwater feedback (Donat-Magnin et al., 2017; Jourdain et al., 2017), which enhances circulation of CDW on the shelf.

Analysis of the heat budget on the continental shelf (Figure 3a) shows that changes in advection are mostly responsible for the warming below 200 m. There is a large ensemble spread, ranging from approximately -20to +700 EJ by the end of the simulation, but the ensemble mean trend is significantly positive at the 99% level. There is a small contribution from vertical diffusion and KPP transport, of  $\approx 9\%$  of the magnitude of the advective trend. Penetration of heat from shortwave radiation has a negligible impact.

A closer look at a representative transect through 106°W (Figure 3b) reveals that advective warming is strongest at middepth (200-500 m), indicating a raised thermocline and stronger onshore flow of CDW (see also Figure S10 in the Supporting Information S1). This is consistent with a wind-driven mechanism, but could also be amplified by the ice shelf meltwater feedback strengthening circulation on the shelf. Vertical diffusion and KPP transport (Figure 3c) partially offsets the changes in advection, and reflects the displacement of the seasonal mixing depth as the thermocline rises. Diffusion and KPP slightly contribute to warming below 500 m, likely related to a reduction in cold convective events that deepen the KPP mixing (Text S4 in the Supporting Information S1). Shortwave penetration (Figure 3d) only affects the near-surface, but a positive trend in this region reflects the loss of sea ice cover, as discussed later.





**a**. Heat budget for continental shelf below 200m

**Figure 3.** Analysis of heat budget terms in the interior ocean: convergence of heat from 3D advection, the sum of vertical diffusion and KPP transport of heat (note there is no explicit horizontal diffusion in this configuration), and penetration of heat from shortwave radiation. Each term is expressed as a time-integrated anomaly from the ensemble mean over the first 30 years, thus representing the accumulation of heat relative to this baseline period. (a) Timeseries (2-year running means) of each term integrated over the Amundsen Sea continental shelf (Figure S1 in the Supporting Information S1) below 200 m, excluding ice shelf cavities. Solid lines show the ensemble mean, while shaded regions show the ensemble spread for each term. "Total" is the sum of the other terms.  $EJ = 10^{18} J$ . (b–e) Ensemble mean trends for each term on the transect through 106°W (Figure S1 in the Supporting Information S1), including Thwaites Ice Shelf. Trends are calculated at each point from annually averaged values, and set to zero where they are not significant at the 95% level. (b and c) share the same color scale. The black dashed line shows the 200 m cutoff. Panel (e) shows the trend in the sum of the displayed terms (b–d), and does not include heat fluxes at the sea surface or the ice shelf base.

The sum of these terms closes the heat budget in the interior ocean (Figure 3e). Total warming is strongest at middepth, which confirms that advective warming resulting in a raised thermocline is the dominant mechanism. There is also advective warming of the deep water below 500 m, which could indicate a stronger onshore flux of CDW below the thermocline. However, it could also be influenced by a reduction in cold convective events,

as quantified in Text S4 in the Supporting Information S1. We find that reduced convection locally amplifies the warming trend close to the coastline, particularly at depth. When convection is accounted for, the underlying warming is pervasive across the shelf, with a similar magnitude in all regions.

#### 3.3. Changes at the Sea Surface

While the warming in our simulations appears to be wind-driven, we cannot determine this unambiguously. Wind stress is not the only forcing which changes during our simulations, and in fact the PACE ensemble exhibits significant trends in multiple atmospheric variables over the last century. Here we examine simulated changes at the sea surface and in sea ice, and discuss their potential relevance to the cavities.

Warming of the near-surface atmosphere ranges from  $\approx 1.2-2^{\circ}$ C/century over the domain (Figure 4a). Surface humidity and precipitation also exhibit positive trends (not shown). These thermodynamic trends can be expected to strengthen in the future as climate change continues. There is also a weakening of the southerly coastal winds (Figure 4b) which drive sea ice export and formation. This trend affects the regions in front of the Abbot, Dotson, and Getz Ice Shelves. There is a similar weakening of the easterly winds in front of Cosgrove Ice Shelf and the westward-facing Abbot ice front (not shown), but no significant trends in Pine Island Bay.

Sea ice freezing weakens (Figure 4c), which is caused by a combination of factors: weakened coastal winds, warmer and moister air, and entrainment of warmer deep water. As less sea ice is exported northward and along the coastal current, sea ice melting also weakens (Figure 4d), particularly near the Getz Ice Shelf. The remaining sea ice is thinner (Figure 4e), especially in regions where ice converges, such as east of Bear Ridge. Sea ice concentration declines almost uniformly, at a rate of  $\approx 10\%$  fractional area per century (not shown). The net surface freshwater trend into the ocean is generally positive (Figure 4f).

Ultimately, the response of the surface ocean is that of warming (Figure 4g) and coastal freshening in Pine Island Bay (Figure 4h), which is mainly driven by the advection of ice shelf meltwater. Warming and freshening of the surface both stratify the water column, which likely contributes to a reduction in convective events. We find that reduced convection can explain some of the warming in our simulations, but is not the main cause (Text S4 in the Supporting Information S1). While our simulations are consistent with a primarily wind-driven explanation, we conclude that a separate study applying idealized forcing changes would be required to determine this unambiguously.

#### 3.4. Discussion and Conclusions

We have presented an ensemble of ice-ocean simulations testing the response of the Amundsen Sea to changes in atmospheric forcing over the last century. We find that prior to the advent of in-situ observations, the Amundsen Sea was likely dominated by cool conditions, with less pronounced warm periods than in recent decades. Our simulations show a clear long-term trend of subsurface warming and increased ice shelf melting, which coincides with glaciological observations of increased ice loss from the WAIS.

The pattern of warming in our simulations, with an increased flux of CDW along established flow pathways onto the continental shelf, is consistent with the expected response to eastward wind trends over the shelf break (Thoma et al., 2008). The link between winds and melting has been examined in decadal observations (Dutrieux et al., 2014; Jenkins et al., 2018), and our simulations confirm this relationship also holds on longer timescales.

It remains a possibility that other mechanisms are contributing to the simulated warming trend. In particular, changes in surface fluxes stratify the water column, which reduces the occurrence of cold convective events. These events have not been observed in a widespread manner in the Amundsen Sea, but it is possible they were commonplace earlier in the century, as our simulations suggest. Further research is needed to clearly separate the roles of westerly winds and surface stratification on Amundsen Sea warming, particularly since the latter mechanism has been found to be important in some future projections of ice shelf melting (Naughten et al., 2018; Timmermann & Hellmer, 2013).

Perhaps most importantly for the wider community, our results build a link between human activity and Amundsen Sea ocean conditions. The influence of Pacific variability, combined with limited oceanographic observations,



## Trends per century in surface variables



**Figure 4.** Ensemble mean trends in surface variables, shown in units per century and calculated using annual means. Trends which are not significant at the 95% level are set to zero, and ice shelves are masked. Sea ice freezing and melting in (c and d) are shown in m/y of freshwater flux into the ocean (positive means freshening; note different color scales), while sea ice thickness in (e) is shown in m of sea ice.

has long obscured the detection of a robust warming trend on the continental shelf. Our simulations provide the missing link connecting long-term wind trends and ice loss from the WAIS. These wind trends are driven by anthropogenic changes in greenhouse gases combined with centennial changes in tropical Pacific variability (P. R. Holland et al., 2019), and are expected to continue in a warming climate (Goyal et al., 2021).

Given that the Amundsen Sea appears to respond to anthropogenic forcing over the last century, it follows that the region should also respond to future anthropogenic forcing. If rising greenhouse gas emissions cause wind trends to continue or accelerate, we can expect the continental shelf to become even more dominated by warm conditions and increased ice shelf melting. Conversely, if emissions are sufficiently reduced and the winds stabilize, further warming of the shelf could be prevented. The future of the Amundsen Sea, therefore, is still to be decided.



#### **Data Availability Statement**

The model source code and configuration files are publicly accessible (Naughten, 2021b); here we use the PAS\_PACE configuration. The pre- and post-processing code has also been publicly archived (Naughten, 2021a). The model output has been archived by the UK Polar Data Centre and is publicly accessible (Naughten, 2022).

#### References

- Asay-Davis, X. S., Jourdain, N. C., & Nakayama, Y. (2017). Developments in simulating and parameterizing interactions between the Southern Ocean and the Antarctic Ice Sheet. Current Climate Change Reports, 3, 316–329. https://doi.org/10.1007/s40641-017-0071-0
- Assmann, K. M., Jenkins, A., Shoosmith, D. R., Walker, D. P., Jacobs, S. S., & Nicholls, K. W. (2013). Variability of Circumpolar Deep Water transport onto the Amundsen Sea continental shelf through a shelf break trough. *Journal of Geophysical Research: Oceans*, 118, 6603–6620. https://doi.org/10.1002/2013JC008871
- Bett, D. T., Holland, P. R., Naveira Garabato, A. C., Jenkins, A., Dutrieux, P., Kimura, S., & Fleming, A. (2020). The impact of the Amundsen Sea freshwater balance on ocean melting of the West Antarctic Ice Sheet. *Journal of Geophysical Research: Oceans*, 125, e2020JC016305. https://doi.org/10.1029/2020JC016305
- Copernicus Climate Change Service. (2017). ERA5: Fifth generation of ECMWF atmospheric reanalyses of the global climate. Retrieved from https://cds.climate.copernicus.eu/cdsapp
- Donat-Magnin, M., Jourdain, N. C., Spence, P., Le Sommer, J., Gallée, H., & Durand, G. (2017). Ice-shelf melt response to changing winds and glacier dynamics in the Amundsen Sea sector, Antarctica. *Journal of Geophysical Research: Oceans*, 122, 10206–10224. https://doi. org/10.1002/2017JC013059
- Dutrieux, P., De Rydt, J., Jenkins, A., Holland, P. R., Ha, H. K., Lee, S. H., et al. (2014). Strong sensitivity of Pine Island ice-shelf melting to climatic variability. *Science*, 343, 174–178. https://doi.org/10.1126/science.1244341
- Fretwell, P., Pritchard, H. D., Vaughan, D. G., Bamber, J. L., Barrand, N. E., Bell, R., & Zirizzotti, A. (2013). Bedmap2: Improved ice bed, surface and thickness datasets for Antarctica. *The Cryosphere*, 7, 375–393. https://doi.org/10.5194/tc-7-375-2013
- Goyal, R., Gupta, A. S., Jucker, M., & England, M. H. (2021). Historical and projected changes in the Southern Hemisphere surface westerlies. *Geophysical Research Letters*, 48, e2020GL090849. https://doi.org/10.1029/2020gl090849
- Gudmundsson, G. H., Krug, J., Durand, G., Favier, L., & Gagliardini, O. (2012). The stability of grounding lines on retrograde slopes. *The Cryosphere*, 6, 1497–1505. https://doi.org/10.5194/tc-6-1497-2012
- Holland, P. R., Bracegirdle, T. J., Dutrieux, P., Jenkins, A., & Steig, E. J. (2019). West Antarctic ice loss influenced by internal climate variability and anthropogenic forcing. *Nature Geoscience*, 12, 718–724. https://doi.org/10.1038/s41561-019-0420-9
- Jacobs, S. S., & Hellmer, H. H. (1996). Antarctic Ice Sheet melting in the Southeast Pacific. Geophysical Research Letters, 23, 957–960. https:// doi.org/10.1029/96GL00723
- Jenkins, A., Dutrieux, P., Jacobs, S. S., McPhail, S. D., Perrett, J. R., Webb, A. T., & White, D. (2010). Observations beneath Pine Island Glacier in West Antarctica and implications for its retreat. *Nature Geoscience*, 3, 468–472. https://doi.org/10.1038/ngeo890
- Jenkins, A., Dutrieux, P., Jacobs, S. S., Steig, E. J., Gudmundsson, G. H., Smith, J., & Heywood, K. J. (2016). Decadal ocean forcing and Antarctic Ice Sheet response: Lessons from the Amundsen Sea. Oceanography, 29(4), 106–117. https://doi.org/10.5670/oceanog.2016.103
- Jenkins, A., Shoosmith, D., Dutrieux, P., Jacobs, S., Kim, T. W., Lee, S. H., & Stammerjohn, S. (2018). West Antarctic Ice Sheet retreat in the Amundsen Sea driven by decadal oceanic variability. *Nature Geoscience*, 11, 733–738. https://doi.org/10.1038/s41561-018-0207-4
- Jourdain, N. C., Mathiot, P., Merino, N., Durand, G., Le Sommer, J., Spence, P., & Madec, G. (2017). Ocean circulation and sea-ice thinning induced by melting ice shelves in the Amundsen Sea. *Journal of Geophysical Research: Oceans*, 122, 2550–2573. https://doi. org/10.1002/2016JC012509
- Kimura, S., Jenkins, A., Regan, H., Holland, P. R., Assmann, K. M., Whitt, D. B., & Dutrieux, P. (2017). Oceanographic controls on the variability of ice-shelf basal melting and circulation of glacial meltwater in the Amundsen Sea Embayment, Antarctica. *Journal of Geophysical Research: Oceans*, 122, 10131–10155. https://doi.org/10.1002/2017JC012926
- Konrad, H., Gilbert, L., Cornford, S. L., Payne, A., Hogg, A., Muir, A., & Shepherd, A. (2017). Uneven onset and pace of ice-dynamical imbalance in the Amundsen Sea Embayment, West Antarctica. *Geophysical Research Letters*, 44, 910–918. https://doi.org/10.1002/2016GL070733
- Losch, M. (2008). Modeling ice shelf cavities in a z coordinate ocean general circulation model. *Journal of Geophysical Research*, 113, C08043. https://doi.org/10.1029/2007JC004368
- MacGregor, J. A., Catania, G. A., Markowski, M. S., & Andrews, A. G. (2012). Widespread rifting and retreat of ice-shelf margins in the eastern Amundsen Sea Embayment between 1972 and 2011. *Journal of Glaciology*, 58, 458–466. https://doi.org/10.3189/2012JoG11J262
- Marshall, J., Hill, C., Perelman, L., & Adcroft, A. (1997). Hydrostatic, quasi-hydrostatic, and nonhydrostatic ocean modeling. Journal of Geophysical Research, 102, 5733–5752. https://doi.org/10.1029/96JC02776
- Nakayama, Y., Menemenlis, D., Zhang, H., Schodlok, M., & Rignot, E. (2018). Origin of Circumpolar Deep Water intruding onto the Amundsen and Bellingshausen Sea continental shelves. *Nature Communications*, 9, 3403. https://doi.org/10.1038/s41467-018-05813-1
- Naughten, K. A. (2021a). knaughten/mitgcm\_python: Pre- and post-processing of PAS configuration (Version v1.0). https://doi.org/10.5281/ zenodo.4892596
- Naughten, K. A. (2021b). knaughten/UaMITgcm: UaMITgcm PAS configuration (Version v1.1). https://doi.org/10.5281/zenodo.4892592
- Naughten, K. A. (2022). Anundsen Sea MITgcm model output forced with Pacific Pacemaker Ensemble, 1920-2013 (Version 1.0). NERC EDS UK Polar Data Centre. https://doi.org/10.5285/A4EA4D64-169A-4981-A64D-C2604B52522E
- Naughten, K. A., Meissner, K. J., Galton-Fenzi, B. K., England, M. H., Timmermann, R., & Hellmer, H. H. (2018). Future projections of Antarctic ice shelf melting based on CMIP5 scenarios. *Journal of Climate*, 31, 5243–5261. https://doi.org/10.1175/JCLI-D-17-0854.1
- Paolo, F. S., Fricker, H. A., & Padman, L. (2015). Volume loss from Antarctic ice shelves is accelerating. *Science*, 348(6232), 327–331. https://doi.org/10.1126/science.aaa0940
- Rignot, E. J., Mouginot, J., Morlighem, M., Seroussi, H., & Scheuchl, B. (2014). Widespread, rapid grounding line retreat of Pine Island, Thwaites, Smith, and Kohler glaciers, West Antarctica, from 1992 to 2011. *Geophysical Research Letters*, 41, 3502–3509. https://doi. org/10.1002/2014GL060140
- Sallée, J.-B., Shuckburgh, E., Bruneau, N., Meijers, A. J. S., Bracegirdle, T. J., Wang, Z., & Roy, T. (2013). Assessment of Southern Ocean water mass circulation and characteristics in CMIP5 models: Historical bias and forcing response. *Journal of Geophysical Research: Oceans, 118*, 1830–1844. https://doi.org/10.1002/jgrc.20135

#### Acknowledgments

During the development of this manuscript, the authors benefited from group discussions with Alberto Naveira Garabato, Alessandro Silvano, Kofan Lu, Oana Dragomir, Jan De Rydt, and Christopher Bull. Martin Losch provided advice while tuning MITgcm. The authors are grateful to the CESM Climate Variability and Change Working Group for producing the Pacific Pacemaker Ensemble and assisting with data access. Computational resources were provided by the ARCHER UK National Supercomputing Service, the NEXCS HPC Service, and the JASMIN data analysis environment. This research is part of the UKRI-JSPS project "Quantifying Human Influence on Ocean Melting of the West Antarctic Ice Sheet" NE/S011994/1.

- Schmidtko, S., Heywood, K. J., Thompson, A. F., & Aoki, S. (2014). Multidecadal warming of Antarctic waters. Science, 346(6214), 1227–1232. https://doi.org/10.1126/science.1256117
- Schneider, D. P., & Deser, C. (2018). Tropically driven and externally forced patterns of Antarctic sea ice change: Reconciling observed and modeled trends. *Climate Dynamics*, 50, 4599–4618. https://doi.org/10.1007/s00382-017-3893-5
- Schneider, D. P., & Steig, E. J. (2008). Ice cores record significant 1940s Antarctic warmth related to tropical climate variability. Proceedings of the National Academy of Sciences, 105, 12154–12158. https://doi.org/10.1073/pnas.0803627105
- Schoof, C. (2007). Ice sheet grounding line dynamics: Steady states, stability, and hysteresis. Journal of Geophysical Research: Earth Surface, 112, F03S28. https://doi.org/10.1029/2006JF000664
- Shepherd, A. P., Ivins, E. R., Rignot, E. J., Smith, B., van den Broeke, M. R., Velicogna, I., & Whitehouse, P. L. (2018). Mass balance of the Antarctic Ice Sheet from 1992 to 2017. *Nature*, 558, 219–222. https://doi.org/10.1098/rsta.2006.1792
- Smith, J. A., Andersen, T. J., Shortt, M., Gaffney, A. M., Truffer, M., Stanton, T. P., & Vaughan, D. G. (2017). Sub-ice-shelf sediments record history of twentieth-century retreat of Pine Island Glacier. *Nature*, 541, 77–80. https://doi.org/10.1038/nature20136
- Spence, P., Holmes, R. M., Hogg, A. M., Griffies, S. M., Stewart, K. D., & England, M. H. (2017). Localized rapid warming of West Antarctic subsurface waters by remote winds. *Nature Climate Change*, 7, 595–603. https://doi.org/10.1038/nclimate3335
- Steig, E. J., Ding, Q., White, J. W., Küttel, M., Rupper, S. B., Neumann, T. A., & Korotkikh, E. (2013). Recent climate and ice-sheet changes in West Antarctica compared with the past 2,000 years. *Nature Geoscience*, 6, 372–375. https://doi.org/10.1038/ngeo1778
- Swart, N. C., Fyfe, J. C., Gillett, N., & Marshall, G. J. (2015). Comparing trends in the Southern Annular Mode and surface westerly jet. Journal of Climate, 28(22), 8840–8859. https://doi.org/10.1175/JCLI-D-15-0334.1
- Thoma, M., Jenkins, A., Holland, D. M., & Jacobs, S. S. (2008). Modelling Circumpolar Deep Water intrusions on the Amundsen Sea continental shelf, Antarctica. *Geophysical Research Letters*, 35, L18602. https://doi.org/10.1029/2008GL034939
- Timmermann, R., & Hellmer, H. H. (2013). Southern Ocean warming and increased ice shelf basal melting in the twenty-first and twenty-second centuries based on coupled ice-ocean finite-element modelling. Ocean Dynamics, 63, 1011–1026. https://doi.org/10.1007/s10236-013-0642-0
- Walker, D. P., Jenkins, A., Assmann, K. M., Shoosmith, D. R., & Brandon, M. A. (2013). Oceanographic observations at the shelf break of the Amundsen Sea, Antarctica. Journal of Geophysical Research: Oceans, 118, 2906–2918. https://doi.org/10.1002/jgrc.20212

#### **References From the Supporting Information**

- Adusumilli, S., Fricker, H. A., Medley, B., Padman, L., & Siegfried, M. R. (2020). Interannual variations in meltwater input to the Southern Ocean from Antarctic ice shelves. *Nature Geoscience*, 13, 616–620. https://doi.org/10.1038/s41561-020-0616-z
- Heywood, K. J., Biddle, L. C., Boehme, L., Fedak, M., Dutrieux, P., Jenkins, A., & Webber, B. G. (2016). Between the devil and the deep blue sea: The role of the Amundsen Sea continental shelf in exchanges between ocean and ice shelves. *Oceanography*, 29, 118–129. https://doi. org/10.5670/oceanog.2016.104
- Holland, D. M., & Jenkins, A. (1999). Modeling thermodynamic ice-ocean interactions at the base of an ice shelf. Journal of Physical Oceanography, 29, 1787–1800. https://doi.org/10.1175/1520-0485(1999)029<1787:MTIOIA>2.0.CO;2
- Jacobs, S., Giulivi, C., Dutrieux, P., Rignot, E., Nitsche, F., & Mouginot, J. (2013). Getz Ice Shelf melting response to changes in ocean forcing. Journal of Geophysical Research: Oceans, 118, 4152–4168. https://doi.org/10.1002/jgrc.20298
- Jenkins, A. (1999). The impact of melting ice on ocean waters. Journal of Physical Oceanography, 29, 2370–2381. https://doi. org/10.1175/1520-0485(1999)029<2370:TIOMIO>2.0.CO;2
- Lenaerts, J. T., Vizcaino, M., Fyke, J., van Kampenhout, L., & van den Broeke, M. R. (2016). Present-day and future Antarctic Ice Sheet climate and surface mass balance in the Community Earth System Model. *Climate Dynamics*, 47, 1367–1381. https://doi.org/10.1007/ s00382-015-2907-4
- Locarnini, R. A., Mishonov, A. V., Antonov, J. I., Boyer, T. P., Garcia, H. E., Baranova, O. K., & Seidov, D. (2013). World Ocean Atlas 2013. Vol. 1: Temperature (Technical Report).
- Losch, M., Menemenlis, D., Campin, J. M., Heimbach, P., & Hill, C. (2010). On the formulation of sea-ice models. Part 1: Effects of different solver implementations and parameterizations. *Ocean Modelling*, 33, 129–144. https://doi.org/10.1016/j.ocemod.2009.12.008
- Meier, W., Fetterrer, F., Savoie, M., Mallory, S., Duerr, R., & Stroeve, J. (2017). NOAA/NSIDC climate data record of passive microwave sea ice concentration, Version 3. NSIDC: National Snow and Ice Data Center. https://doi.org/10.7265/N59P2ZTG
- Merino, N., Le Sommer, J., Durand, G., Jourdain, N. C., Madec, G., Mathiot, P., & Tournadre, J. (2016). Antarctic icebergs melt over the Southern Ocean: Climatology and impact on sea ice. Ocean Modelling, 104, 99–110. https://doi.org/10.1016/j.ocemod.2016.05.001
- Morlighem, M. (2020). *MEaSUREs BedMachine Antarctica, Version 2*. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/E1QL9HFQ7A8M
- Morlighem, M., Rignot, E., Binder, T., Blankenship, D., Drews, R., Eagles, G., & Smith, E. C. (2020). Deep glacial troughs and stabilizing ridges ice sheet. *Nature Geoscience*, 13, 132–137. https://doi.org/10.1038/s41561-019-0510-8
- Naughten, K. A., Meissner, K. J., Galton-Fenzi, B. K., England, M. H., Timmermann, R., Hellmer, H. H., & Debernard, J. B. (2018). Intercomparison of Antarctic ice-shelf, ocean, and sea-ice interactions simulated by MetROMS-iceshelf and FESOM 1.4. Geoscientific Model Development, 11, 1257–1292. https://doi.org/10.5194/gmd-11-1257-2018
- Naughten, K. A., De Rydt, J., Rosier, S. H., Jenkins, A., Holland, P. R., & Ridley, J. K. (2021). Two-timescale response of a large Antarctic ice shelf to climate change. *Nature Communications*, 12, 1991. https://doi.org/10.1038/s41467-021-22259-0
- Peng, G., Meier, W. N., Scott, D. J., & Savoie, M. H. (2013). A long-term and reproducible passive microwave sea ice concentration data record for climate studies and monitoring. *Earth System Science Data*, 5, 311–318. https://doi.org/10.5194/essd-5-311-2013
- Verdy, A., & Mazloff, M. R. (2017). A data assimilating model for estimating Southern Ocean biogeochemistry. Journal of Geophysical Research: Oceans, 122, 6968–6988. https://doi.org/10.1002/2016JC012650
- Zhang, J., & Hibler, W. (1997). On an efficient numerical method for modeling sea ice dynamics. *Journal of Geophysical Research*, 102, 8691–8702. https://doi.org/10.1029/96JC03744
- Zweng, M. M., Reagan, J. R., Antonov, J. I., Mishonov, A. V., Boyer, T. P., Garcia, H. E., & Bidlle, M. M. (2013). World Ocean Atlas 2013, Volume 2: Salinity (Technical Report).